Author's response to reviews

Title: Design of the PRINCESS Trial: Pre-hospital Resuscitation Intra-nasal Cooling Effectiveness Survival Study (PRINCESS)

Authors:

- Per Nordberg (per.nordberg@sodersjukhuset.se)
- Fabio S Taccone (ftaccone@ulb.ac.be)
- Maaret Castren (maaret.castren@sodersjukhuset.se)
- Anatolij Truhlar (atruhlar@seznam.cz)
- Didier Desruelles (didier.desruelles@uzleuven.be)
- Sune Forsberg (sune.forsberg@sodersjukhuset.se)
- Jacob Hollenberg (Jacob.Hollenberg@ki.se)
- Jean-Louis Vincent (jlvincen@ulb.ac.be)
- Leif Svensson (leif.svensson@sodersjukhuset.se)

Version: 4 Date: 16 November 2013

Author's response to reviews:

Dear Editor, we have added:

- names of Ethics Committees and reference numbers as requested
- a document stating the peer-reviewing process for funding.

Kind regards
Fabio Silvio TACCONE